Author’s response to reviews

Title: Prevalence of multimorbidity in the elderly population: results of a study based on electronic medical records

Authors:

JOSE M ABAD (jmabad@aragon.es)
AMAIA CALDERON (acalderon.iacs@aragon.es)
ANTONIO PONCEL (aponcel@salud.aragon.es)
BEATRIZ POBLADOR (bpoblador.iacs@aragon.es)
JOSE M CALDERON (jmcalderon@aragon.es)
ANTONI SICRAS (asicras@bsa.cat)
MERCEDES CLERENCIA (mclerencia@salud.aragon.es)
ALEXANDRA PRADOS (sprados.iacs@aragon.es)

Version: 3 Date: 3 October 2013

Author’s response to reviews: see over
Author's response to proposal of changes from the editors

**Title:** Prevalence of multimorbidity in the elderly population: results of a study based on electronic medical records

**Authors:** JOSE M ABAD, AMAIA CALDERON, ANTONIO PONCEL, BEATRIZ POBLADOR, JOSE M CALDERON, ANTONI SICRAS, MERCEDES CLERENCIA and ALEXANDRA PRADOS

**Version:** 2

**Date:** 3 October 2013

**Author's response:** see over
Research article

Prevalence of multimorbidity in the elderly population: results of a study based on electronic medical records

JOSE M ABAD, AMAIA CALDERON, ANTONIO PONCEL, BEATRIZ POBLADOR, JOSE M CALDERON, ANTONI SICRAS, MERCEDES CLERENCIA and ALEXANDRA PRADOS

BMC Health Services Research

Comments from the editors

We have now looked over your manuscript and we feel that it is out of scope for the journal.

However, we do feel that the manuscript may be suited for another of our journals, BMC Geriatrics (http://www.biomedcentral.com/bmcgeriatr).

If you would be happy to have your manuscript transferred to this journal for further consideration, please let us know within the next seven days and we will handle the transfer on your behalf. Your manuscript will then be assessed by an Editor and sent for peer review if appropriate.

Response from the authors

We have no problem with the transfer of the manuscript to BMC Geriatrics, so you have our approval to proceed with the transfer.

Comments from the editors

Moreover, before we can proceed with peer review we will need you to make some changes to the formatting of your manuscript. We strongly urge you to make these changes promptly, as we cannot start the peer review process until we have received a version containing the changes.

*Acknowledgements

Please acknowledge anyone who contributed towards the article by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include the source(s) of funding for each author, and for the manuscript preparation. Authors must describe the role of
the funding body, if any, in design, in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication. Please also acknowledge anyone who contributed materials essential for the study. If a language editor has made significant revision of the manuscript, we recommend that you acknowledge the editor by name, where possible.

Response form the authors

We have modified the manuscript according to your suggestions, and we have included an Acknowledgements section with information about the source of funding for the research in a revised version of the manuscript:

ACKNOWLEDGEMENTS

The study was funded by Grant PI11/01126 from the Carlos III Health Institute. The funding body had no role in the study design, data collection, data analysis, data interpretation, or decision to publish the findings.